Author's response to reviews

Title: Phenotypic and Genetic Characterization of a Cohort of Pediatric Wilson Disease Patients

Authors:

Tawhida Y Abdel Ghaffar (tyghaffar@gmail.com)
Solaf M Elsayed (elsayed683@yahoo.com)
Suzan Elnaghy (suzan_elnaghi@yahoo.com)
Ahmed Shadeed (Ahshadeed@hotmail.com)
Ezzat S Elsobky (elsobkyezzat@hotmail.com)
Hartmut Schmidt (hepar@ukmuenster.de)

Version: 6 Date: 5 June 2011

Author's response to reviews: see over
June 6, 2011

Dear editor,

Please find enclosed the revised manuscript titled “Phenotypic and Genetic Characterization of a Cohort of Pediatric Wilson Disease Patients” that we submit as an original manuscript to publish in your journal.

Some additional typing errors were corrected

Sincerely yours,

Tawhida Y. Abdel Ghaffar, Solaf M. Elsayed, Suzan Elnaghy, Ahmed Shadeed, Ezzat S. Elsobky, Hartmut Schmidt

**Corresponding Author**

Tawhida Y. Abdel Ghaffar

**Address:** Yassin Abdel Ghaffar Charity Center for Liver disease and research. 6 Emarat El Tasniea St. from Makram Abied, Nasr city, Cairo, Egypt

**e-mail:** tyghaffar@gmail.com

**telephone:** 202-22720525- +20122285619

**Fax:** 202-22733806